Author's response to reviews

Title: Cerebral misery perfusion diagnosed using hypercapnic BOLD-contrast fMRI: a case report

Authors:

  Adam L Gordon (adam.gordon@nottingham.ac.uk)
  Stephen Goode (msxsg@nottingham.ac.uk)
  Olympio D'Souza (olympiodsouza@yahoo.com)
  Dorothee P Auer (dorothee.auer@nottingham.ac.uk)
  Sunil K Munshi (sunil.munshi@nuh.nhs.uk)

Version: 6 Date: 14 January 2010

Author's response to reviews:

Revisions are as requested in your recent email:

- title in the submission software and the manuscript now match. Both are in sentence case.
- abstract in the submission software and the manuscript now match.
- Dorothee P Auer's middle initial has been added in the submission system.
- Job titles (by which I assume you mean the prefixes Dr and Prof) have been removed from the manuscript.
- Underlined text has been removed from the manuscript.
- the vertical line has been removed from our table.
- a full list of authors has now been provided for reference 7.
- I have checked the typography of the article and am happy with it.